Author's response to reviews

Title: Predictors of overweight and obesity in adult women in Nairobi Province, Kenya

Authors:

Regina W Mbochi (re.mbochi@yahoo.co.uk)
Elizabeth Kuria (enkuriakan@yahoo.com)
Judith Kimiywe (sjokimiywe@yahoo.com)
Sophie Ochola (socholasa55@gmail.com)
Nelia P Steyn (npsteyn@hsrc.ac.za)

Version: 2 Date: 4 July 2012

Author's response to reviews: see over
Dear Sir, attached please find a revised manuscript for publication in BMC Public Health entitled: Predictors of overweight and obesity in adult women in Nairobi Province, Kenya

We have tried to address all the reviewers’ comments as well as those made by the editor. I believe the manuscript has been improved as a result.

Sincerely,

Dr Nelia Steyn RD (SA), MPH, PhD
Chief Research Specialist
Human Sciences Research Council
Private Bag X9182, Cape Town, 8000
12h Floor, Plein Park Building, 69-83 Plein Street, Cape Town, 8001
Tel: +27 21- 466 7832
Fax: +27 21- 466 8069
Email: npssteyn@hsrc.ac.za
Web: www.hsrc.ac.za